Author’s response to reviews

Title: Neuroprotective effects of Eucommia ulmoides Oliv. and its Bioactive Constituent Might Work via Ameliorating Ubiquitin Proteasome System

Authors:

Hong Guo (cacti1983@163.com)
Fang Shi (sflrq@163.com)
Meijiao Li (meijiaojiao851020@163.com)
Qingqing Liu (liuqingqing8913@126.com)
Bin Yu (yubin_771115@hotmail.com)
Limin Hu (huliminth@126.com)

Version: 4 Date: 17 February 2015

Author’s response to reviews:

Dear editor,

This is a revision of the previous manuscript submitted in BMC Complementary and Alternative Medicine entitled “Neuroprotective effects of Eucommiae Ulmoides Oliv. and its chemical compounds for treatment of Parkinson’s disease via ameliorating ubiquitin proteasome system” (ID: 1235502227142721). Thank you for your letter and for the reviewers’ comments concerning our manuscript. We have studied the comments carefully and have made correction that we hope meet with approval. The changes were marked-up by the function of amendment in the revision.